Author's response to reviews

Title: Favorable Prognostic Value of SOCS2 and IGF-I in Breast Cancer

Authors:

Michael C Haffner (Michael.Haffner@i-med.ac.at)
Barbara Petridou (Barbara.Petridou@jouy.inra.fr)
Jean Phillipe Peyrat (JP-Peyrat@o-lambret.fr)
Francoise Revillion (f-revillion@o-lambret.fr)
Elisabeth Muller-Holzner (elisabeth.mueller@i-med.ac.at)
Gunter Daxenbichler (guenter.daxenbichler@i-med.ac.at)
Christian Marth (christian.marth@i-med.ac.at)
Wolfgang Doppler (wolfgang.doppler@i-med.ac.at)

Version: 2 Date: 3 July 2007

Author's response to reviews: see over
July 3rd, 2007

To the Editor of BMC Cancer

Dear Editor,

We would like to resubmit our manuscript on “Favorable prognostic value of SOCS2 and IGF-I in breast cancer” by Haffner et al. to BMC Cancer. In response to the criticisms of reviewer 2 we have extended the description of the RT-PCR technique on p4 of the Methods section and have included representative original data as Suppl. Fig. 1. We also have added on p 12 in the Discussion part a reference to the paper of Yamashita. The abstract was changed to accommodate for the style of BMC Cancer and statements of Competing interests and Authors’ contribution were added.

I hope that with these changes the manuscript is suitable for publication in BMC Cancer.

Yours sincerely,

Wolfgang Doppler, M.D.Ph.D.